



Article Explainable Ontology-Based Intelligent Decision Support System for Business Model Design and Sustainability

Basma Hamrouni ^{1,*,†}, Abdelhabib Bourouis ^{2,†}, Ahmed Korichi ^{1,†} and Mohsen Brahmi ^{3,†}

- Computer Science Department, University of Ouargla, Ouargla 30000, Algeria; ahmed.korichi@gmail.com
 Research Laboratory on Computer Science's Complex Systems, University of Oum El Bouaghi,
 - Oum El Bouaghi 04000, Algeria; bourouis.abdelhabib@univ-oeb.dz
- ³ Faculty FEM, University of Sfax, Sfax 3018, Tunisia; brahmi.mohsen@gmail.com
- Correspondence: besma.hamrouni1@gmail.com
- + These authors contributed equally to this work.

Abstract: Background: Case-Based Reasoning (CBR) is a problem-solving paradigm that uses knowledge of relevant past experiences (cases) to interpret or solve new problems. CBR systems allow generating explanations easily, as they typically organize and represent knowledge in a way that makes it possible to reason about and thereby generate explanations. An improvement of this paradigm is ontology-based CBR, an approach that combines, in the form of formal ontologies, case-specific knowledge with domain one in order to improve the effectiveness and explanation capability of the system. Intelligent systems make daily activities more easily, efficiently, and represent a real support for sustainable economic development. On the one hand, they improve efficiency, productivity, and quality, and, on the other hand, can reduce costs and cut waste. In this way, intelligent systems facilitate sustainable development, economic growth, societal progress, and improve efficiency. Aim: In this vision, the purpose of this paper is to propose a new generation of intelligent decision support systems for Business Model having the ability to provide explanations to increase confidence in proposed solutions. Findings/result: The performance results obtained show the benefits of the proposed solution with different requirements of an explanatory decision support system. Consequently, applying this paradigm for software tools of business model development will make a great promise for supporting business model design, sustainability, and innovation.

Keywords: decision support system; business model; case-based reasoning; sustainability; explanation; ontology

1. Introduction

Sustainability is fast becoming a need that has to be seriously confronted by decisionmakers, enterprises, and consumers [1,2]. With the growing human population, environmental and social degradation is getting more severe and our options are diminishing [3,4]. However, technological advances towards sustainability are increasing rapidly, and many companies find it difficult to meet their sustainability goals. Therefore, to leverage sustainable solutions innovation on the business model level, it is indispensable to align incentives and revenue mechanisms [3,5].

Today, decision-making requires taking into account an increasing amount of data, information, and knowledge of different kinds and qualities in the various activities of the company; competitiveness depends on their analysis and optimal exploitation. As a result, managers are increasingly using Decision Support Systems (DSS), which are interactive computer-based systems planned to help decision-makers using communications technologies, data, documents, knowledge, and models to identify and solve problems, complete decision process tasks, and help make decisions; this is for the purpose to enhance a person or a group's ability to make decisions [6]. Decision support methods are designed to support decision-making processes concerning an extensive range of problems. When



Citation: Hamrouni, B.; Bourouis, A.; Korichi, A.; Brahmi, M. Explainable Ontology-Based Intelligent Decision Support System for Business Model Design and Sustainability. *Sustainability* **2021**, *13*, 9819. https:// doi.org/10.3390/su13179819

Academic Editor: Antonino Galati

Received: 31 July 2021 Accepted: 25 August 2021 Published: 1 September 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). the challenge is the analysis of many often conflicting decision criteria [7–9], multicriteria decision analysis (MCDA) methods are used to help making the choice of the best compromise [7–9]. The Multicriteria Decision Aid (MCDA) is a field of the operational research discipline that manipulates complex decision-making problems handling high uncertainty and conflicting objectives [10]. Recently, the integration of MCDA tools into DSS has provided decision-makers with powerful capabilities in analyzing, exploring, and comparing a set of alternatives. Many studies have presented Multicriteria Decision Support Systems [7,11,12]. In fact, this paper suggests Intelligent Decision support systems that are the result of the fusion of Artificial Intelligence (AI) with DSSs. The intelligent decision support systems (IDSS) has helped in widening the window of current research and application in information processing and analysis. As a result, the systems are smarter, quite efficient, adaptable, and better able to aid human decision-making. The IDSSs are part of the AI with the main objective of filling the gap between decision-making tools and human interactions [13].

The power of Information and Communications Technology (ICT) is increased by AI and its constitutive elements of data, algorithms, hardware, connectivity, and storage, and this is a major opportunity for Sustainable Development. Recently, ICT branch is making the world's energy infrastructure more efficient, that is demonstrated in the Global e-Sustainability Initiative (GeSI) 2008. This initiative concluded that buildings and transport, smart grids, along with travel substitution could reduce global carbon emissions by a net 15% and save up to €600 billion by 2020 [14].

In 2020, the GeSI Smarter 2020 study found that the total abatement potential of ICTenabled solutions in 2020 was about 9.1 gigatons of carbon dioxide equivalent (GtCO2e), a saving of ~16.5% of global GHG emissions by 2020. This is a saving of 21.6 billion barrels of oil and approximately equivalent as USD 1.9 trillion in gross energy and fuel savings [15,16].

In another view, digitization and dematerialization, depending fundamentally on existing technologies, substitute the need for a carbon-intensive product achieved 0.5 GtCO2e. Research mentioned that the reduction of emissions is 1.5 GtCO2e from the use of social media and networking (data collection and communication) and intelligent simulation, the automation of infrastructure, and industrial processes more broadly were shown to save 4.7 GtCO2e, while Systems integration—primarily building or industrial management systems and the use of less-carbon intensive, renewable energy technologies—saved 2.4 GtCO2e [16].

Artificial Intelligence has received increased interest from academic scholars. Therefore, it has to be understood as the ability of a system to act intelligently and to do so in ever-wider regions [17,18]. The arrival of AI is modifying a number of sectors, and it is expected to affect global productivity [19], environmental outcomes [20], and other areas, both in the short- and long-term. The authors of [21] argued that AI can enable the accomplishment of 134 targets across all the United Nations Sustainable Development Goals (SDGs).

The aim is to provide and manage intelligent products, services, and experiences through the sharing of information for cooperation or creation of optimal and sustainable value [22]. This branch of computer science can influence production and consumption patterns to achieve sustainable resource management according to SDGs outlined in the UN 2030 Agenda [18]. The growing of AI and the significant potential of development in the different sectors of society determine the evaluation of its effects on sustainable development [21]. For this reason, companies have to face the challenge of sustainability by improving innovations to preserve the use of natural resources and the integrity of the ecosystem [23–25].

In the past few years, the business model literature gave little importance to the social and environmental challenges facing the world today. The concentration was on market sustainability instead of social and environmental sustainability [26]. For instance, the authors of [27] argued that the "understanding of sustainable business models and how

sustainable development is operationalized in firms is weak" [28]. Consequently, recent researches argued that developing sustainable business models is a strong way to combine the benefits of using business models and to search for the performance at the financial, social, and environmental levels. To integrate sustainability at their business, managers need instruments to perform change in their current business models [26]. This question of how managers can reflect on their business model has gained more interest in recent years [29]. Therefore, companies should benefit from existing methodologies developed over recent decades in operations research, complex systems analysis, and artificial intelligence. These techniques are increasingly used to develop sustainable solutions [30].

Many researchers have contributed to the area of business model design [31–33], innovation [34,35], and start-ups BMs [25]. They concluded that additional tools are needed to support the design and evaluation of BM, which has led to the development of computer-based tools. This potential is reflected not only in recent calls for business model development tools (BMDT) in the Information System (IS) discipline, but also in adjunct disciplines [36], in which the introduction of software-based tools has already successfully contributed to an improvement, among others, of the development of strategies and products. Software-based tools for strategy making enable easily to integrate and document information from various sources during the process of ideating and progressing such strategies [37].

While software-based BMDTs are said to have great potential to support their users in innovating BMs [38–40], especially with the use of AI techniques as machine learning [41], little attention has been given to providing highly accurate decisional guidance by creating trust in these AI-based DSSs [41]. The capability of an intelligent system to provide a meaningful explanation of its actions is a crucial factor affecting the acceptance of the system [42]. Consequently, this explanation is important to enable end users to understand, trust, and effectively manage their systems. Furthermore, the BMDTs have to focus on providing assistance to the user in being creative, which was neglected by existing BM design tools. The author of [37] suggested that BMDTs could incorporate functions for exploring business model patterns. The author of [43] has argued that the challenges of BM-related education involve the limits to creativity imposed by existing frameworks and the difficulty in prototyping new business models. The solution is to combine explanation and experience-based learning approaches into a virtuous learning cycle. Given such explanation capability, the user is more likely to accept the decisions, suggestions, or results provided by the system.

Therefore, with regard to business applications, the use of the BM ontology in an ontology-based CBR approach can be regarded as the next logical step. Unfortunately, no significant attention has thus far been paid to ontology-based CBR concerning the inclusion of BM ontologies. For this reason, this paper proposes an intelligent decision support for business models that offer highly accurate decisional guidance into account by integrating explanations for designing and innovating BMs, with a case-based reasoning approach that uses domain knowledge from the BM domain.

The remainder of this paper is organized as follows. Section 2 presents key concepts and definitions. Section 3 outlines literature on related works. Section 4 introduces the overall proposed system and main contributions of this paper and discusses the architecture of the proposed solution. Section 5 is devoted to the CBR process. Section 6 explains the implementation of the proposed solution, and its performance evaluation is depicted in Section 7. The last section concludes the paper and gives some perspectives for future enhancements of this solution.

2. Background

2.1. Decision Support Systems

Many types of computer-based information systems (IS) have been developed to support decision-making, including decision support systems [44]. DSSs were first developed in the 1970s, and have been used widely since the personal computer revolution in

the 1980s. DSSs can be described as "computer-based systems that help decision-makers confront ill-structured problems through direct interaction with data and analysis models" [45]. They are designed to increase the speed and accuracy of data analysis, while reducing costs, enabling the effective and efficient analysis of large volumes of quantitative data. They were originally developed as tools for managers, but they are now also used by many non-management employees such as sales people and purchasing officers. They are particularly valuable tools in complex situations, where decision-makers need to analyze multiple sources of data [44].

Decision support methods are designed to support decision-making processes concerning an extensive range of problems. When the challenge is the analysis of many often conflicting decision criteria [7–9], multicriteria decision analysis (MCDA) methods are used to help the decision to make the choice of the best compromise [7–9]. The Multicriteria Decision Aid (MCDA) is a field of the operational research discipline that manipulates complex decision-making problems handling high uncertainty and conflicting objectives [10]. Recently, the integration of the MCDA tools into DSS has provided decision-makers with powerful capabilities in analyzing, exploring, and comparing a set of alternatives, many studies have presented Multicriteria Decision Support Systems [7,11,12].

Multicriteria Decision Analysis (MCDA) methods use an approach in which an expert in a given field defines the model, and then the decision variants are evaluated [10,13]. The author of [11] proposed a decision-making model based on already evaluated alternatives and stochastic optimization techniques, without the need to expert's interfering with the process. Classic methods of multicriteria decision analysis are used to analyze alternatives described by using numerical values, to solve the problem in decision-making methods in which the decision-maker makes decisions based on partially incomplete data. The author of [7] proposed a fuzzy models with partially incomplete data and in [12] proposed a technique for selecting and evaluating suppliers in an incomplete fuzzy using the fuzzy MCDM method, that provide more acceptable and efficient outcomes to select the best alternatives without the affection by the experts' inclinations.

The fusion of AI with DSSs has helped in widening the window of current research and application in information processing and analysis. As a result, the systems are smarter, quite efficient, adaptable, and better able to aid human decision-making. In order to develop systems, the most usable possible, in the 1990s, DSSs were enriched by techniques rooted in AI, particularly the introduction of a knowledge base into their architecture, so as to give the system the capacity for reasoning [46]. The literature survey in [47] presents the broad historical progression of DSS research. A number of DSSs were presented such as Model-driven DSS and Data-driven DSS, then the emergence of Communication-Driven DSS, Web-based DSS, and Knowledge-driven or Knowledge Based DSS (KB-DSS) that can suggest actions to the manager. The authors of [48] discussed recent advances, challenges, and evolution perspectives of KB-DSS.

2.2. Case-Based Reasoning

Case-based reasoning is a common methodology for humans to solve problems. Therefore, this methodology is well applied in business, while people do not know that they use this approach by default. Scholars highlighted that the case-based reasoning methodology, implemented in information systems, can be successfully applied in business [49]. It is an adequate methodology for experience management to support people who have to perform complex and knowledge intensive tasks. Case-based reasoning (CBR) is an AI approach that solves a new problem by remembering a previous similar situation and by reusing information and knowledge of that situation [50]. CBR is based on the human cognitive model and is meant to mimic our way to think, predict, and comprehend. It has been used in many successful applications due to its psychological plausibility [51]. The principal idea of CBR systems is to store the experiences of people in the knowledge base as cases. When there is a new problem, we can search the knowledge base, finding the result of the past similar problem and the process of problem solution. Then, we can have an acceptable case as a reference. If the user is unsatisfied with the case, it can adjust the case to suit the current situation. Finally, this revised case will be stored in the case base [52]. CBR can be considered as a five-step problem solving process:

- 1. Case representation: a description of the current problem is input to the system;
- 2. Case retrieval: the system retrieves (similar) past cases from the case base relevant to solving the current problem.
- 3. Case adaptation: the system uses solution used for the previous case to generate target situation to the current problem.
- 4. Case Revising: the solution is validated through feedback from the environment or the user. It is tested in the real-world or using simulation.
- 5. Case Retaining: if appropriate, the validated solution is added to the case base for use in future problem solving.

The retrieve step consists of subtasks such as treat identifying features, initial match, search, and select [50]. Choosing indices is a critical step to CBR system, and three types of case indexing are used. In nearest neighbor approach, the retrieval system searches similar cases based on a weighted sum of features between cases. The inductive approach is based on the most important features. It indexes past cases affecting the outcome as induced from the data itself. In knowledge-guided approach, existing domain and experimental knowledge is applied to locate relevant cases [53]. Guidelines suggested for selecting indexing features include predictiveness, abstractness, concreteness, and usefulness [54].

2.3. Explanation of System Decisions

Furthermore, it is commonly known that in everyday human–human interactions, explanations are an important vehicle to convey information in order to understand one another. Explanations enhance the knowledge of the communication partners in such a way that they accept certain statements and allowing them to make informed decisions. It has long been recognized that increased user understanding of system reasoning may increase user confidence in an intelligent system's decisions [55]. CBR systems cases provide a natural vehicle for explanation. Unlike more opaque methods such as neural networks, CBR systems present the cases on which decisions are based, in this way the outputs are intuitively explained. Users can then compare the current situation to the prior cases to assess their applicability. Some CBR work considered the case as a sufficient explanation; recent work is developing a richer view [56]. In addition to providing arguments for the relevance of a case and a conclusion in the form of comparisons and contrasts [57], other research examines the facets affecting system conclusions, such as how features of a case contribute to similarity calculations to fulfill explanation [58].

2.4. Business Model

Today, research on business models is a rapidly growing field that is still in search of a strong theoretical foundation. The business model concept can be broadly defined as an abstract representation of business logic [59] that supports practitioners in conceiving, designing, and communicating business ideas, and serving as a reference framework [40,60,61]. The business model literature finds its roots in many research branches such as strategic management [62], technology and innovation management [63], or information systems [60]. Business models are considered by many as purposefully designed systems focus on identifying principles, patterns, elements, or configurations of successful business models [52].

IT Support Software and tools that support BM management have already been developed but are still in the earliest state of immaturity and are largely restricted to the design phase, by supporting the visualization of a BM [40,61]. However, recent research highlights that such "computer-aided business modeling tools should go beyond simple design tools and evolve into an own class of high-level decision support tools" [39,61]. This implies that software tools not only support design, but the overall process. Business model knowledge was represented by informal text [40,64], structured text [40,65], morphological

representation [66], ad hoc graphical representation [67], conceptual models with defined semantics and dedicated graphical representations [33], and ontologies [32,40].

3. Related Work

Various research works have been carried out on information systems. In this paper, we are interested in research devoted to approaches used for DSSs. We focus more particularly our review on existing studies that aim at integrating explanation in DSSs and satisfying the design requirements. The reading of the literature review of our problematic case reveals that research in this area has taken a big step in the DSS domain. However, only few studies have focused on intelligent decision support system for the Business Model. To our knowledge, there is no DSS for BM that have the ability of providing explanations. In this section, we limit our review to the existing studies adopting the CBR approach and DSS having the ability to provide explanations, as our proposed solution combines these concepts.

3.1. DSS for Business Model

The first stream of BMDT scholars in the context of BM design and innovation focuses on modeling languages [37]. These languages facilitate analysis, communication, and documentation of business model ideas, especially using graphical notations to represent the basic logic and the most important elements of a business model. The "e3-value" language was proposed in [33]. It allows to capture and evaluate BMs from a financial point of view. The authors of [68] presented a toolbox design that offers a set of soft and hard methods for designing BMs. Business Model Canvas is a visual chart for developing new or documenting existing BMs [69]. Strategic Business Model Ontology (SBMO) is defined as a layer on top of the goal-oriented modeling framework. The authors of [70] proposed a framework composed principally of the strategic business model ontology (SBMO) for representing and analyzing business model canvas to a novel form of ontology that contain more in-depth information on a company's Business Model [71].

Further research focused on the use of analogies for product or process innovations to stimulate the generation of business model ideas, and researches noted that most "new" business models are a recombination of existing ideas, concepts, and patterns [72]. The authors of [73,74], generate BM in the form of catalogs of business model examples. Business model patterns was proposed in [72]. A set of four tools were developed by the INSPIRE project related to business model archetypes, which uses 55 Business Model patterns and serves as an inspirational tool to consider innovative value propositions and revenue models [75]. Scholars highlighted that these IT tools supporting the process of designing, innovating, and evaluating a company's business model are currently not leveraging the full potential of tool support [38].

Another research stream focused on validating the BM with analytical methods such as using modeling and simulation [60,76–78]. Simulations provide a time-efficient and cost-efficient way to help decision makers understand the consequences of business model adaptions without endangering an organization [79]. In [71,80,81], quantitative and qualitative scenarios analysis was applied to predict the viability of design decisions in the context of business model innovation. Other researches were interested by stochastic analysis of financial models to identify the most important drivers of financial performance [78].

A third research stream scholars used Artificial Intelligence techniques as machine learning with collective intelligence to validate the BM [41]. However, little attention has been given to provide highly accurate decisional guidance by creating trust in these AI-based DSSs [41,82]. If explanations about the decision process are given, the user is more likely to accept the decisions and understand the solution provided by the system. In addition, understanding the reasons behind decisions of black box decision systems is nowadays a crucial topic [83].

3.2. DSS and Explanation

During the last two years, the awareness for explanation-aware computing increased rapidly as a result of the growing call for providing an explanation to support the decision made [84]. The capability of a knowledge-based system to provide a meaningful explanation of its actions is a crucial factor affecting its acceptance [42]. Since the earliest promises of computing and artificial intelligence, the explanation power of intelligent systems has attracted growing interest in recent decades. In the late 1970s and 1980s, expert systems using knowledge bases [85] and production rules [86] were the first to integrate the explanation dimension, especially for medical decisions. Rule extraction was developed to interpret artificial neural networks (ANN) and support vector machines (SVM) from the 1990s to 2000s [87,88].

For recommender systems in the 2000s, researches showed that the acceptance of the recommendations was increased with explaining to the user why a recommendation was made [89–91]. Recent research focused on explanation methods for Bayesian Belief Networks [92], ontology-based explanation for ANN [93], and explanation in Case-Based Reasoning systems [56].

Explanation in Expert Systems, Bayesian Networks, and Case-Based Reasoning influenced later research on explaining intelligent systems. Furthermore, the recent interest in interpretable Machine Learning is not well connected to this body of work [91].

AI techniques and algorithms which generate explanations are characterized by the way they support human reasoning and how to represent explanations with visualization methods, data structures, and atomic elements. Similarity-based modeling technique seeks to group similar objects and identify distinguishing features to differentiate between objects. This technique uses the same method as people learning general concepts [94]. Various AI approaches have been developed, including similarity modeling with distance-based methods such as case-based reasoning [50] and clustering models [95], classification into different kinds as in supervised models [96], nearest neighbors [97], dimensionality reduction to find latent relationships such as collaborative filtering [98], principal components analysis [99], and matrix factorization [100]. Many of these methods are data-driven to match candidate objects with previously seen data (training set), where characterization depends on the features engineered and the model which frames an assumed structure of the concepts. Explanations of these mechanisms are driven by inductive and analogical reasoning to understand why certain objects are considered [94].

The growing field of eXplainable AI (XAI) in recent years was driven by DARPA's initiative to fund XAI in the 1970s [101]. There has been interest in explanations of intelligent systems with expert systems [85,102], Bayesian networks (for a review, refer to the work in [87]) and artificial neural networks in the 1980s [87], and recommender systems in the 2000s [89,90]. Recently, AI and machine learning are achieving major successes for several applications. However, machine learning algorithms need to be able to explain how they arrive at their decisions. Consequently, there has been increased attention into interpretable and eXplainable AI.

3.3. DSS and Case-Based Reasoning

Case-based reasoning (CBR) is a nature-inspired paradigm of machine learning capable of continuously learning from past experiences [103]. It uses already solved problems to solve a new one, and its working principle is reasoning by remembering. This principle implies that the reasoning used to solve a target problem is remembered. Thanks to the advantages offered by the CBR method, such as being able to deal with data with noise, missing data, unstructured data, and being closer to the human decision-making process than rule-based systems, CBR applications are various in almost every field of knowledge. In addition to medical diagnosis [104], e-commerce [105], bankruptcy prediction [106], scheduling and process planning [107,108], customer classification [109], fault diagnosis [110,111], prediction of information system outsourcing success [112], concur-

rent product design [113], risk analysis [114], knowledge management [115], and military control [110], CBR can also be applied to mine big data [116].

The CBR system can use three types of ontologies [117]:

- 1. Domain ontologies providing the vocabulary for describing a domain and interpreting a description of a problem in that domain,
- 2. Task ontologies providing the vocabulary for describing terms involved in the problem solving processes,
- 3. Common sense ontologies including a wide range of foundational knowledge as time, space, or causality [118].

To the best of our knowledge, CBR systems have not been used in DSS for Business Model. However, CBR is well adapted to this context because it leads itself easily to reasoning about context and situations by storing and remembering specific episodes (situations), and it has been suggested as a candidate for computational reasoning about situations [119]. Due to the creativity and diversity of Business Model cases and the lack of associated knowledge retrieving and intelligent systems, this paper aims at the design and implementation of a CBR system for Business Model design and validation. The structure of CBR systems allows the user to consider the relation between cases, which is important for an explanation [120]. The CBR technique is based on examples and a means of assessing the similarity or differences between examples. Consequently, the explanation of the recommendation given by the decision support system is done by presenting similar cases that motivate the recommendation.

In summary, there were three major limitations in the previous studies:

- 1. The few proposed DSSs for BM do not assist the user in the first idea of the project.
- 2. The DSSs for BM lack of explanation about how the system reaches the solution. Based on the preliminary review on the various scholars, this study aimed to develop a CBR-based decision support system for the design and validation of the BM.
- 3. Scholars indicated that such IT-based guidance might be perceived as missing the in-depth support of personal mentors [41]. For that, the proposed knowledge based system includes all in-depth knowledge about BM in the form of ontology.

3.4. Intelligent DSS and Sustainable Development

For the social impact, AI enables all the targets by supporting the provision of water, food, health, and energy services to the population. Moreover, it supports low-carbon systems, for instance, by underpinning the creation of circular economies and smart cities that efficiently use their resources [121,122]. AI integrates variable renewables by facilitating smart grids that partially match electrical demand to times when the sun is shining and the wind is blowing [121]. For the economic outcomes of, the AI has positive impact associated to increased productivity and consumption reduction.

Environmental benefits of IA are that with analyzing large-scale interconnected databases enabling to develop joint actions aimed at preserving the environment. AI advances can support the understanding of climate change and the modeling of its possible impacts. Furthermore, it can support low-carbon energy systems with high integration of renewable energy and energy efficiency, which are all needed to address climate change [121,123,124]. Neural networks improve the classification of vegetation cover types based on satellite images, with the possibility of processing large amounts of images in a short time [21].

AI has revolutionized the ways of doing business [125]. It has influenced trade and management practices in number of sectors that offer increasingly competitive and sustainable products or services [18,126]. Indeed, the interaction between artificial technologies and human intelligence is based on algorithms that should help managers make the right decisions, generating a cultural drift in which a large number of data, connections, and interactions become part of the standard management of organizations [125]. They have well-cataloged and organized information sets, so much so that previous research has even

shown that in many situations, these models are more efficient than human decisions [127]. These mathematical models simplify the work of managers [18].

Knowledge Management Systems (KMS) promote the management of intelligent work [128], as they put together organizational, human, and technological factors in the process of creating value through the sharing of knowledge [129]. KMS exploits AI for the improvement of operational processes and business models [130].

From a business model view, sustainable change can be reassured through the renewal of the corporate purpose and organization, oriented by innovation strategies that point toward the creation of long-term value [131,132]. A business model provides a "concise representation of how a related set of decision variables in the areas of business strategy, architecture and economics are addressed to create a sustainable competitive advantage in markets" [133]. A BM draws the boundaries of the operational and physical structure of the company, presaging the consequences of corporate operations in the system architecture, encouraging sustainable choices through innovative strategies for the creation of value [134].

Thus, profitability and sustainability at the business level represent two major objectives that have to be integrated into company's strategy. They are not antagonistic concepts; they can work together in order to develop and improve business models. Moreover, considering the state-of-the-art in the field of business models, during the last 20 years, the BMs were reinvented based on the challenges generated by the international business environment [135].

The use of AI in Sustainable BMs (SBMs) can support managers' choices in decisionmaking and management processes through the use of data to make forecasts and reducing the cost of projections themselves, encouraging a change in the culture and behavior of organizations [136]. Thus, profitability and sustainability at business level represent two major objectives that have to be integrated into company's strategy. They are not antagonistic concepts; they can work together in order to develop and improve business models. Moreover, considering state of the art in the field of business models, during the last 20 years, BMs were reinvented based on the challenges generated by the international business environment [135].

4. The Overall Structure of EIDSS-BM

Most enterprises agree that knowledge is an essential asset for success and survival on an increasingly competitive and global market. This awareness is one of the main reasons for the exponential growth of knowledge management in the past decade. Our approach to BM design is based on ontologies, and makes knowledge assets intelligently accessible to people in organizations, that knowledge resided only in the heads of people.

The system allows decision-makers to assess, understand, measure, change, communicate, or even simulate their business models and allow them to introduce changes and experiment with it in order to learn about business opportunities. The system allows offers a visualization tool of the actual business model of a firm and, on the other hand, equally allows the representation of virtually possible business models. This would facilitate innovation and could also help managers understand the business models of competitors. The past cases of BM are well cataloged and organized that simplify the design of the BM by decision makers. Then, based on the adaptation phase of the CBR system, a new business model could be suggested. By using a number of iterations for learning, managers can do risk-free experiments without endangering their organization.

The system assists decision-makers by providing knowledge about environmental production and reverse logistics processes, hybrid products (integrated and associated services) focusing on sustainability ideas in order to open up new markets for sustainable products, innovations about how companies preserved, and improve the use of natural resources and the integrity of the ecosystem.

This is a pioneering work for creating a case base ontology of Business domain (Business Models) for CBR systems. The initial idea of the Intelligent Decision Support System for Business Models (IDSS-BM) has been discussed in [82].

Our contribution consists of an explainable and Intelligent DSS (EIDSS-BM) for BM design. This is motivated by the fact that unstructured intangible experiences and knowledge are usually difficult to represent and instantiate, which engenders the hardship of knowledge transfer and sharing. This system presents a new paradigm of the CBR based on ontology. It consists of retrieving, reusing, revising, and retaining cases and has been proved effective in retrieving information and knowledge from prior situations. The system handles well the integration of the prediction of the success of the BM to more explanation and guidance for the decision-maker.

Ontology-based CBR is an approach that combines, in the form of formal ontologies, case-specific knowledge with domain one in order to improve the effectiveness and explanation of the system. The authors of [137] combined CBR, Rule-Based Reasoning (RBR), and ontology to develop a solution retrieval system for fault diagnosis. An ontology-based CBR information retrieval system method that integrates CBR and natural language processing (NLP) techniques for metro accident case retrieval has been proposed in [138].

Consequently, case-based reasoning system can benefit from ontology-based knowledge representation. This gives the possibility to reuse existing domain knowledge about business models during the execution of the case-based reasoning cycle. Applying this paradigm to software tools of business model development has great promise for supporting business model design and innovation. Moreover, case semantic retrieval algorithms can be improved by using case-base and domain background knowledge in the form of ontologies, and our approach uses the Business Model Canvas for this purpose.

4.1. Architecture of EIDSS-BM

Many CBR architectures have been presented in the literature [50,139]. Inspired by these architectures, EIDSS-BM architecture is composed of three layers: a data layer, a processing layer, and an application layer as shown in Figure 1.

The data layer contains three main containers: vocabulary, case base, and similarity measures. In knowledge-intensive CBR Systems, ontologies play an important role in representing these containers. They can be used as the vocabulary to describe cases and/or queries, as a knowledge structure where the cases are located, and as the knowledge source to achieve semantic reasoning methods for similarity assessment [140]. The vocabulary and the case base rely on two knowledge models, the domain and case models are respectively presented above.

The processing layer is the core part of the system. The proposed system aims to design BM and aid managers in their decisions. Many types of contextual information cannot be easily inferred [141]. Based on the BM ontology model, the CBR module is used for suggesting the design of BM following four steps which are feature selection, case retrieval, case matching, and case updating. The CBR module is able to obtain the different alternatives of the BM because it exploits historical cases being actually happened.



Figure 1. EIDSS-BM architecture.

4.2. Domain Model

In the CBR system, ontology plays two main roles: the first as case base and the second as domain ontology. The combination of ontologies for domain and case base will achieve the Knowledge-Intensive CBR (KICBR) [31].

The domain model is represented by ontologies which contain the vocabularies, concepts, and relations for representing knowledge concerning Business Models. Following the classification proposed in [142], we have developed two types of ontologies: a core ontology that contains the different parts of the business model in more detail for a deeper understanding such as the value proposition that is the central characteristic of a business model. This ontology provides the formalization of elements, relationships, vocabulary, and semantics of the essential knowledge about BM domain. This ontology is built on BMO illustrated in Figure 2. It contains knowledge about the four main pillars of a BM which are the products and services that a firm offers, the infrastructure and the network of partners, the relationship capital, and the financial aspects [143].



Figure 2. E-Business Model Ontology e-BMO [143].

The value proposition settles the value that is delivered to the customer, which can be a product, a service, or a combination of both [144]. Nevertheless, when choosing which value should be delivered, the customer and provider have to rely on their competences and resources. The organization model analyzes the capabilities or skills and the firms role in the value chain. If the provider is not able to take over the tasks or services he is obliged to by the value proposition, he cannot offer a certain business model. Therefore, value proposition and organization model have to be considered at the same time as they form the backbone of a business model [145].

A domain ontology describes the domain of BM Components. This part of the ontology is built on the BM ontology BMO [32], which is an important contribution to the theory and practice of business models. Its concepts are specializations of other concepts of the core ontology, after collecting information about the business model of startups and technology companies, including attributes and relations of companies, people, and investments. Moreover, the reuse of ontologies from a library also benefits from their reliability and consistency. The more knowledge is embedded into the system, the more effective is expected to be.

4.3. Case Model

The present paper considers BM experience as cases in the sense of case-based reasoning. The CBR framework can be used to retrieve historic BM from a knowledge base, the case base, which is represented as an ontology. It is built on the business model ontology BMO [32] which is an important contributions to the theory and practice of business models.

The proposed BM-ontology includes the important terminology used for the representation of cases in CBR systems according to the methodology described in [118]. A case, in general, is described by a pair (problem, solution) containing problem attributes and solution attributes. Accordingly, the enterprise description part of our model corresponds to the problem part, and the BM part corresponds to the solution part. Therefore, the two main parts are a description part describing the context and the enterprise and a BM part describing the different possible values of the components of a BM. Inspired from the approach of CBR ONTO [118], in order to enhance the communication between the case base and the domain model, the case model is represented within an ontology that integrates the domain model.

This ontology, illustrated in Figure 3, contains the main following concepts:

- 1. CBR-CASE that subsumes the various case types that may exist in the System;
- 2. CBR-DESCRIPTION that subsumes the case main parts company description (name, number of employees, country), and BM components (value proposition, customer segment, key resources, customer relationship, key activities, cost structure, key partners, revenue stream).
- 3. CBR-INDEX allows to integrate the domain model concepts used to describe cases. The central characteristic of a business model which was identified in the literature is the value proposition. It is analyzed by the majority of scholars that regard business model components [146]. The value proposition settles the value that is delivered to the customer, which can be a product, a service or a combination of both. Nevertheless, when choosing which value should be delivered customer and provider have to rely on their competences and resources [32,145].



Figure 3. The case model ontology.

14 of 28

5. The CBR Process

The CBR, proposed in [147], can simulate human cognitive processes and integrates empirical knowledge of different fields into a unified format [45]. The CBR can achieve effective and accurate BM design. This is because that CBR can obtain and suggest BM components and innovation methods as long as cases are able to be matched successfully in the case base. Furthermore, CBR can also exploit historical cases which have actually happened and have been solved, to ensure accuracy.

The process of CBR-based EIDSS-BM is shown in Figure 4. Industry, Year of foundation, Number of employees, and Country are selected as feature indexes for calculating case similarities in the case base. The project is selected as an essential feature if case matching is successful then the BM components of the matched case are displayed directly. Then case updating is evaluated, and if successful then the case base will be updated, else the CBR process will end. When the similarity value is less than the desired value, the system will fail and the process will end.



Figure 4. The CBR Process in EIDSS-BM.

5.1. Case Retrieval

Case retrieval refers to searching the business model case base based on Industry, Country, and Year of foundation. The system uses ontologies as a persistent media for the cases. The ontology defines the type and values of the attribute of the case. User defines the problem by entering information regarding Industry, Country and Year of foundation. The case retrieval engine module takes the created semantic query vector generated and searches for the most similar *k* cases in the case base ontology. The solution space consists

of a list of alternatives of business models for the current project type, results and benefits with each of them.

5.2. Case Matching

The nearest neighbor algorithm was used to calculate the similarity value [46]. Its formula is given in Equation (1).

$$Sim(C_i) = \sum_{j=1}^{m} (w_j * Sim(C_{ij}))$$
(1)

where *i* is the sequence number of the case in the case base. $Sim(C_i)$ is the similarity value between the target case and the *i*th case. *j* is the sequence number of the feature index. *m* represents the numbers of all feature indexes. w_j is the weight of the *j*th feature index. $Sim(C_{ij})$ is the similarity value between the *j*th feature of the *i*th case and the target case.

The threshold of $Sim(C_i)$ is set as Sim(C), which means that the case-matching is invalid when $Sim(C_i) < Sim(C)$; otherwise, case-matching is successful. The value, w_j , is directly defined by system experts together with domain experts.

5.3. Case Updating

When case matching is achieved successfully, the actual business model results can be saved into the case base.

5.4. Explanation

The ability of any software system is improved by increasing its understandability, which in turn can be supported by appropriate explanation capabilities [120]. We follow Schank [148] in considering explanations the most common method used by humans to support understanding and their decision making. In everyday human–human interactions explanations are an important vehicle to convey information in order to understand one another. Explanations enhance the knowledge of the communication partners in such a way that they accept certain statements. They understand more, allowing them to make informed decisions.

Ease-of-use is of high priority for the development of myCBR applications integrating explanations into the user interface. In order to increase transparency and trust in the retrieval process [149], myCBR creates an explanation object for each case during similarity calculation.

Increased user understanding of system reasoning may increase user confidence in an intelligent system's decisions, thanks to CBR systems cases that provide a natural vehicle for explanation.

In case-based reasoning, conceptual explanations are used to explain the vocabulary knowledge container. Backward explanations make clearer and justify the outcome of a particular retrieval result and provide means for understanding the results of a similarity calculation [150].

5.4.1. Conceptualization Goal

A conceptual explanation is a comprehensive description of a concept. It answers questions about concepts, about terms and concepts of ten arise for the end user when he or she is not familiar with the application domain. It consists of a definition, some examples, and references to further characterizations, for which any kind of medium can be used (e.g., text, images, audio, and video). Conceptual explanations are inherently static, because concepts usually do not change. However, there are good reasons to consider the context in which the concept is used and the user's personal level of knowledge [151].

5.4.2. Transparency Goal

The user could ask how the system reached the conclusion presented, and an explanation in the form of a reasoning trace from the system would be presented. This allows the users to check the system by examining the way it reasons and allows them to look for explanations for why the system has reached a surprising or anomalous result. The use can ask the question "How did the system come to the similarity assessment of a particular case?" when he/she wants to retrace the procedure of similarity assessment, thus achieving the transparency goal.

5.4.3. Justification Goal

The justification goal is closely related to the transparency goal. Where transparency is concerned with presenting the reasoning trace, justification deals with the ability to explain why an answer is good. Justification is often preferable over transparency, as simply displaying the reasoning trace is not always sufficient and can even be counter productive [74,119]. With regard to Explainable Artificial Intelligence (XAI), CBR is particularly interesting because similar cases can be used as examples for justifying the reasoning of the system. This can be considered as an interpretable model.

However, in terms of explanations, most CBR systems are limited to the display of similar cases. Compared to "black box" algorithms such as deep learning, the responses of CBR systems can be justified easily using similar cases as examples. The user can have in mind the question "Which are the most similar aspects of a case? Which are the least?", so the answer explains in which way the case is similar to the query and thus achieving the transparency goal.

6. EIDSS-BM Implementation

6.1. Overview

myCBR is an open-source plug-in for the open-source ontology editor Protégé [152]. Protégé is an extensible Java-based tool that provides a plug-and-play environment making it a flexible base for rapid prototyping and application development [153]. It allows defining classes and attributes in an object-oriented way. Furthermore, it manages instances of these classes, which myCBR interprets as cases [151]. Therefore, the handling of vocabulary and case base is already provided by Protégé. The myCBR plug-in provides several editors to define similarity measures for an ontology and a retrieval interface for testing [154]. The main goal of myCBR is to minimize the effort for building CBR applications that require knowledge-intensive similarity measures.

To develop a CBR system, various programming tools are available both freely and commercially. Among these, myCBR and jCOLIBRI are the most widely used and known frameworks for teaching and academic research purposes. myCBR has a number of features and provides a very easy way to develop case-based reasoning applications. It supports fast prototyping and combining state-of-the-art CBR functionalities. It can work as a standalone and also as a plugin of Protégé.

6.2. Implementation

Three steps are required to develop a CBR application:

- Generation of case representations.
- Modeling similarity measures.
- Testing of retrieval functionality.

Generation of case representations:

One powerful feature provided by myCBR is the easiness of the case representation by CSV data import module [154]. CSV files are widely used to store attribute-value based raw data in pure ASCII format. Using the CSV importer, the user has the choice to import data instances into an existing Protégé data model, or to create a new model automatically based on the raw data. Figure 5 shows how Business Model dataset is arranged in a CSV file. myCBR allows also slots to be added manually using Protégé. Figure 6 shows myCBR screen after importing the dataset into a new class Business Model which will be used as query and case values for retrieval step.

6		(× (≈ -) ∓		Busine	ssModelC	aseBase - Micr	rosoft Excel					_	Ø	×
E	Accu	il Insertion Mise en page Formules	Données B	évision Affichage										σx
C	B×.	Calibri - 11 - A A = =	= »·	Renvoyer à la ligne automatiquement	Standard		55			7	Σ Somm	e automatique *]	1	A
C	oller 🧹	B I ∐ · ⊞ · 🍐 · 📥 · ≡ ≡	= :: :: :	E Fusionner et centrer *	∰ - %	·	Mise en forme M	ettre sous forme	Styles de Insére	r Supprimer For	nat 🖉 Effacer	- T	rier et Rech	ercher et
Pre	sse-pa 🕫	Police 5	A	Vignement G	Nom	nbre 🦻	on an on the ne	Style	centres	Cellules		Édition	inter seree	aoniner
-	B8 Coursera						*							
							1							-
	A	В	С	D		E	F	G	Н	1	J		K	-
1	case id	Name	Industries	site		Description	Founded	Employees	CRCLASS	Customer Re	VPCLASS	Value propos	ition	
2		1 826Valencia	Education	http://826valencia.org/		Business Mo	2002	100000		Published wo	ork of the stu	After-school	tutoring: 0	Jne-
3		2 Amazon Kindle Direct Publishi	iş ketali	http://kdp.amazon.com		Business Mo	1995	100000		Independent	sen-publishi	Independent	sen-publ	Isninj
4		3 Alphabet (Google)	Tanual	www.google.com		Business Mo	1998	10000		Interaction w	Online 8 m	web search e	engine, ca	lenda
6		4 Airbito	IT	https://www.airbrib.com		Business Mo	2008	10000		Transactiona	Long torm	The world's l	able plat	rinor
7		6 Alexies Pharmacouticals	IT.	http://www.businessinoueiganery.co	om /	ionhormooo	1002	10000		Derconalized	i,cong term	Life transform	ingest bus	luete
0		7 Coursers	Education	https://www.alexionpharma.co	onny	Social entror	2012	10000	Collaborativ	Community	Accossibility	Access to wo	declare e	ucts,
9		8 Lidemy	Education	https://www.coursera.org/		Business Mo	2012	1000	conaborative	Transactiona	I	Access to a la	rae numb	ner of
10		9 WyzAnt	Education	http://www.wuzant.com		WyzAnt is an	2005	100		On the one h	, and short-te	Fasy way to f	ind a tuto	r at a
11		10 SweetBush	Education	http://www.sweetrush.com/		Business Mo	2001	1000		on the one h		Custom conte	ent develo	opme
12		11 Kiwi Crate	Education	http://www.kiwicrate.com		Kiwi Crate cr	2011	100		Kiwi Crate fa	Accessibility	Delivers fun h	ands-on	expe
13		12 Meine Spielzeugkiste	Education	https://www.meinespielzeugk	kiste.de/	Meine Spielz	2012	100		Transactiona	1	Access to a la	rge variet	vof
14		13 Trinity School	Education	www.trinityschoolnyc.org		Business Mo	1709	1000		Community,	Accessibility	nspiring & ca	ring teach	iers,C
15		14 University of the People	Education	http://uopeople.org/		Provides acc	2009	100	Collaborative	Community,	Accessibility	High-quality a	and low-c	ost h
16		15 Western Governors University	Education	http://www.wgu.edu/		WGU is a pri	1997	10000	Collaborative	Self-instruction	Accessibility	Competency-	based ed	ucati
17		16 Alibaba Group	IT_Hardwar	ehttp://www.alibaba.com/		The world's	1999	50000	Unbundling_	Free email, e	Accessibility	Simple & safe	trade so	lutio
18		17 Amazon Web Services	IT_Hardwar	ehttp://aws.amazon.com/		Amazon.com	1955	100000		Long term rel	lationships th	Broad & deep	core clo	ud in
19		18 McDonald's				The leading	f 1940	100000		Website,Soci	al Media,Mc	Fast & easy fo	od,Quali	ty sta
20		19 McKinsey & Co.	Consulting_	Phttp://www.mckinsey.com/		McKinsey &	1926	50000		Long term, of	ten persona	Consulting: in	dividual,	judge
21		20 ScribeAmerica	Training	http://www.scribeamerica.co	m	Business Mo	2004	10000		Personal,Lon	g-term,Comr	Reduces the	workload	of pł
22		21 The Lego Group	Learning	http://www.lego.com		One of the la	1932	50000		One of the la	rgest toymak	To inspire an	d develop	the 💌
H	< > > E	usinessModelCaseBase 🖓												

Figure 5. Business Model Dataset in a CSV File.

≪ EIDSS-BM Protégé 3.5 (file:\C:\Program%20F	iles%20(x86)\Protege_3.5\%20EIDSS-BM.	pprj, Protégé Files (.pont and .pins))	- □	×
Eile Edit Project Code Window Collaboration	on Tools Help				
	4 4 V			I pro	tégé
😑 Classes 🐂 Slots 📑 Forms 🔶 Instan	ces 🛛 🚢 Queries 🖌 \varTheta Explanation Edit	or 🦷 🥮 CBR Retri	eval 🤍 🥮 Similarity Measure Editor		
CLASS BROWSER	CLASS EDITOR				
For Project: 🖶 EIDSS-BM	For Class: O Business Model (instan	ce of :STANDARD-C	LASS)	X (× ×
0 - 1 × × -				0.14	
Class Hierarchy 🛛 🐴 🍷 🗮 🗶 🎽	Name		Documentation	Constraints 💏 🔭 🕈	
THING	Business Model				
V O SYSTEM-CLASS					
ANNOTATION	Role				
CONSTRAINT	Abstract O				
META-CLASS					
RELATION	Template Slots			요 요 美 前 =	* =: _
Business Model	Name	Cardinality	Туре	Other Facets	
	Channels	single	String		-
	CHCLASS	single	Integer	minimum=2147483647, maximum=0	
	 company type 	single	Symbol	allowed-values={Established,Startup,B2B,Not_for_profit,Established,B2B}	
	COSCLASS	single	Integer	minimum=2147483647, maximum=0	
	Cost Structure	single	String		
	Country	single	Symbol	allowed-values={United_States,Germany,Denmark}	
	CRCLASS	single	Symbol	allowed-values={Collaborative_Production,Unbundling_Bundling}	100
	CSCLASS	single	Integer	minimum=2147483647, maximum=0	
	 Customer Relationships 	single	String		
	Customer Segments	single	String		
	 Description 	single	String		
	Employees	single	String		
V 00	Founded	single	String		
8	 How does the business model work 	single	String		
	 Industries 	single	Symbol	allowed-values={IT,Hardware,Software,Education,Retail,Travel,Pharmaceuticals,Li	fe
STOTEWPOD-00	KACLASS	single	Symbol	allowed-values=Horizontal_Expertise	
	Key activities	single	String		
	Key partners	sinale	String		-



Modeling similarity measures:

After having generated the case representation by using the CSV importer, the main task for creating a CBR application with myCBR is the definition of an appropriate similarity measure. Here, myCBR follows the local-global approach which divides the similarity definition into a set of local similarity measures for each attribute, a set of weights, and a global similarity measure for calculating the final similarity value. Figure 7 shows one of the implemented similarity measures.

There are several similarity modes, to choose between depending on the type of the slot. In our example, the slots values of the industry can be arranged in a hierarchical form and the taxonomy as similarity mode was used.

The resultant structure specifies parent–child relations through the position of the objects in the taxonomy. This means that a node has at least the same attributes its parent node has. For instance, "Learning" is also "education" and the real-world sectors are "learning" and "training". Therefore, "education" is an abbreviation for all companies with "industries" being "Learning" or "Training". This choice enhances the retrieval results.

 EIDSS-BM Protégé 3.5 (file:\C:\Prog Eila Edit Project Code Window (gram%20Files%20(x86)\Protege_	3.5\%20EIDSS-BM.pprj, Protégé Files (.pont and .pins))		- 🗆 X
	: ≥ ≤ 4 4 4				< protégé
🔍 Classes 👘 Slots 📑 Forms	🕈 Instances 🛛 🚢 Queries	Explanation Editor BR Retri	eval 🤅 🥮 Similari	y Measure Editor	
CLASS BROWSER	SIMILARITY MEASURE EDIT	DR			
For Project: • EIDSS-BM	Available Functions:				
Class Hierarchy 🦂 🎽 🍝 🗶 🛨	default	New	Duplicate	"default" is the active similarity measure	
	Similarity mode: Taxonomy Symmetry: Symmetric	• I asymmetric I			
Business Model State	Inner nodes as values	Query: O no O yes Case: O no O yes	i	Industries Carl Life science Pharmaceuticals Carl During Augusteenary sector [0.6]	
SLOT BROWSER For class ● Business Model Slot Hierarchy & 🗧 💥 🗙	Semantic of inner nodes	Query: O any_value Image: O any_value O any_value Image: O uncertain Image: O uncertain	I	Consulting_ProfessionalServices	
 How does the business model work Industries KACLASS Key activities 	Semantic of uncertain	Image: Constraint of the system Image: Constraint of the system Image: Constraint of the system Image: Constraint of the system Image: Constraint of the system Image: Constraint of the system Image: Constraint of the system Image: Constraint of the system	I		
Key partners Key Resources KPCLASS KPCLASS			+	Ca Travel	
Name Revenue Streams Revenue Streams			4 1000	a	

Figure 7. Similarity editors for symbolic attributes with taxonomy editor.

Testing of retrieval functionality:

Figure 8 gives a schematic overview of the Retrieval GUI. The names of the classes' attributes are presented with the rows of the table. The area is divided into several columns. The left most column is used for query specification. The others are used to show the retrieval results. The rightmost column lists all cases of the case base ordered by their similarity to the query.

EIDSS-BM Protégé 3.5 (file:\C\Program%20Files%20(x86)\Protege_3.5\%20EIDSS-BM.pprj, Protégé Files (pont and .pins))							
Elle Edit Project Code Window Collaboration Tools Help							
					protégé		
🖉 Channes 🖉 Challer 💭 Frances 🕼 Austrian Caller 🧳 Colonation Caller 🌑 Colonation Caller							
	Explanation Earlor	Crai Commany incasare caro		OUEDV DECULTS			
DETAILS AND ODERY QUERY RESULTS							
Business M.,. V - Retrieve P Load P Save X Clear - Res	et		< Previous > Next	2 EIDES BM_Class14	0.97		
				2 EIDES BM_Class ID	0.97		
Query	EIDSS-BM_Class14	EIDSS-BM_Class15	EIDSS-BM_Class7	A FIDSS-BM_Class11	0.97		
(autory)	1 🕀	2 🕀	3 🕀	5 EIDSS-BM_Class13	0.96		
	0.97	0.97	0.97	6 EIDSS-BM_Class10	0.93		
Action oriented insights Coursera is one of the most widely	The business model of the Univers	1. Innovating traditional models to	Coursera is one of the most widely	7 EIDSS-BM Class20	0.93		
Annual revenue		297.56 million U.S. Dollar		8 EIDSS-BM Class9	0.93		
CHCLASS undefined	undefined	undefined	undefined	9 EIDSS-BM Class1	0.9		
COSCLASS undefined	undefined	undefined	undefined	10 EIDSS-BM Class8	0.9		
CRCLASS Collaborative Production	Collaborative Production	Collaborative Production	Collaborative Production	11 EIDSS-BM Class17	0.76		
CSCLASS undefined	undefined	undefined	undefined	12 EIDSS-BM Class19	0.76		
Channels Online platform, Mobile application,	Online,Word of mouth	Online, Bricks and mortar universit	Online platform, Mobile application,	13 EIDSS-BM Class12	0.73		
Cost Structure Creation of IT infrastructure,Busine.	.IT infrastructure,Human resources	Fees for course materials, Operatio.	Creation of IT infrastructure, Busin	14 EIDSS-BM Class21	0.73		
Country United States	United States	United States	United States	15 EIDSS-BM Class16	0.67		
Customer Relationships Community, Social media¶.Blog	Community, Guided classes	Self-instruction, Guidance through	Community, Social media¶, Blog	16 EIDSS-BM Class18	n.a.		
 Customer Segments Mass market, Individuals interested. 	Individuals who are at least 18 yea	Working adults (without degree) w	Mass market, Individuals intereste	17 EIDSS-BM Class3	0.6		
Description Social entrepreneurship company	Provides access to undergraduate	WGU is a private, nonprofit univer	Social entrepreneurship company 💈	18 EIDSS-BM Class5	0.6		
Employees 1000	100	10000	1000	19 FIDSS-BM Class6	0.6		
Founded 2012	2009	1997	2012	20 EIDSS-BM Class2	0		
 How does the business model work undefined 	University of the People is the worl.	Western Governors University (W	undefined	21 EIDSS-BM Class4	0		
Industries Education	Education	Education	Education				
KACLASS undefined	undefined	undefined	undefined				
KPCLASS undefined	undefined	undefined	undefined				
KRCLASS undefined	undefined	undefined	undefined				
Key Resources Online platform, Partnerships with t	Technological resources, Global In	Competency-based educational m	Online platform, Partnerships with t				
Key activities Creating & maintaining IT infrastru	Creating and maintaining the IT inf	Development of competencies, Pro	Creating & maintaining IT infrastru				
Key partners Universities & teachers, Amazon, Tr.	Corporations, Foundations, Academ	Licensing partners (e.g., Pearson,	Universities & teachers, Amazon, T				
Name Coursera	University of the People	Western Governors University	Coursera				
RSCLASS undefined	undefined	undefined	undefined				
Revenue Streams Premium services: Signature Trac	Processing fees for application an	Tuition fees, Program-specific fees,	Premium services: Signature Trac	Start: 6:05:17			
VPCLASS Accessibility	Accessibility	Accessibility	Accessibility	Finish: 6:05:17			
Value proposition Access to world-class education fo.	High-quality and low-cost higher e	Competency-based educational m	Access to world-class education f	Duration: 0,003 sec			

Figure 8. Retrieval of a case query.

myCBR includes an easy to use GUI for performing retrievals and for analyzing the corresponding results. By providing similarity highlighting and explanation functionality, after the initial knowledge model was created, a number of retrieval experiments using the myCBR were performed. We tested the 22 records that are in the case base and only one missed case is obtained. Figure 8 shows one query of these records after retrieving the most similar cases.

Enhanced retrieve using ontology:

A domain-specific ontology or user-defined ontology represents specific domain knowledge or relation between words using class and subclass [155]. Terms of a case can be compared with other cases by exact matching. However, some slots like the "industries" of the company can have a relationship with other synonyms or industries. The weight of

such terms can be increased automatically using a domain specific ontology defined by experts. For example, the term "education" is related to "learning" and "training", and it is more similar to some industries, so that enhance the result of the user query.

6.3. Implementation of Conceptual Explanations

Conceptual explanations are addressing the system end users. They are interested in retrieving the most similar cases to their queries. At the bottom of the retrieval GUI, conceptual explanations are shown when the user hovers the mouse over a table cell. Figure 9 shows a snapshot. Conceptual explanations do not involve complicated algorithms. The required functionality at its core is a static mapping from concepts to explanations.



Figure 9. Conceptual explanations.

The figure displays all concepts of the ontology. On the right-hand side, one can edit its explanation like a short description and a list of URLs of further documents. For example, the "start up" in companies can be explained via Wikipedia or Google define.

6.4. Implementation of Backward Explanations

6.4.1. Implementation of Transparency Goal

For a query, the system delivers a ranking of the case base. However, the retrieval result may be quite surprising and need some explanation. To answer this question "How did the system come to the similarity assessment of a particular case?" The system offers a detailed recording of the retrieval process. Therefore, every step of similarity calculation can be traced. To increase transparency, myCBR creates an explanation object for each case during the retrieval. This tree-like data structure stores global and local similarity values as comments for each attribute.

6.4.2. Implementation of Justification Goal

The answer of the question "Which are the most similar aspects of a case? Which are the least?" The notion of aspect must be clarified, an aspect is one single attribute. The similarity of an aspect is then a local similarity value and can be displayed with myCBR.

7. Testing and Evaluation

After implementing the proposed EIDSS-BM, its performance has been evaluated. The purpose of the evaluation process is to get the end user's views on the significance and usefulness of the system.

The primary data used in this research are based on the dataset of the "Business Model Gallery" [156] that is a publicly available database on business models, as a resource for learning about different BM designs. In this database, BMs from all kinds of industries are explained allowing to learn about various patterns. This project aspires to capture the power of analogies by being a place for inspiration but this system do not use the case-based reasoning. It provides a variety of details including the company information about the industry, employees, foundation, country, company type, the annual revenue, and the different values of the BM components.

7.1. Performance Metrics

We use the following performance metrics for the evaluation of the system:

NCases: The number of cases contained in the case base over time is very important. It is desirable to enrich the case base while preserving or even increasing precision.

As in [157], for effectiveness, we examined the relevance of the retrieved cases given a test case. The retrieval effectiveness can be defined in terms of *precision* and *recall* rates.

Precision: The percent of retrieved cases that are similar to the new case ($N_{Correct}$) among the total number of retrieved cases ($N_{Correct} + N_{false}$).

Recall: The percent of retrieved cases by the system, which are similar to the new case $(N_{Correct})$, among the total number of cases actually similar to the new case in the case base (N_{Total}) .

$$precision = \frac{N_{Correct}}{N_{Correct} + N_{false}}$$
(2)

$$recall = \frac{N_{Correct}}{N_{Total}}$$
(3)

where $N_{Total} = N_{Correct} + N_{Missed}$ denotes the total number of tagged cases that are similar to the new case and $N_{Correct}$ denotes the number of retrieved cases that are similar to the new case. N_{false} is the number of retrieved cases that are dissimilar to the new case and N_{Missed} is the number of tagged cases that are similar to the new case but not retrieved.

7.2. Comparison between the EIDSS-BM and the Business Model Gallery

The system's performance has been evaluated in terms of *precision* and *recall*, which are two common metrics used to estimate the efficiency of the information retrieval in a CBR system. *Precision* is the fraction of a searched output that is relevant to a particular query. It represents the proportion of the relevant retrieved cases to all the retrieved cases and the less irrelevant cases are retrieved the better the precision will be. Therefore, the calculation requires knowledge of the relevant and non-relevant hits in the evaluation set of cases. *Recall*, on the other hand, measures the ability of a retrieval system to obtain all or most of the relevant cases from a collection of cases (case base). It represents the proportion of relevant cases are retrieved to all the relevant cases in the case base. The more relevant cases are retrieved the better the recall requires knowledge not just for the relevant and retrieved cases but also those are not retrieved yet relevant.

To evaluate the correctness of the retrieval function, the CBR retrieval test is performed. For that test, relevant BM cases (from the case base) should be identified for each test case. For identification of relevant cases, test cases are given to the domain expert in order to assign possible relevant cases (from the case base) for each one. Then, the precision and recall are evaluated. Table 1 summarizes the results of this process where test cases are {Case 7, Case 40, Case 51 and Case 62} and for each one, the set of relevant cases identified by the expert (from the case base).

Test Case	Relevant Cases from the Case Base
Case 7	Case 9, Case 10, Case 8
Case 40	Case 18, Case 22, Case 7, Case 8, Case 9, Case 10, Case 11, Case 12, Case 13, Case 14, Case 15, Case 20, Case 21, Case 3, Case 5, Case 6, Case 16, Case 17, Case 19
Case 51	Case 53, Case 54, Case 61, Case 68, Case 75, Case 81
Case 62	Case 55, Case 57, Case 69, Case 72, Case 71, Case 59

Table 1. Relevant cases assigned by domain experts for sample test case.

Once the relevant cases are identified and assigned to the test cases, the next step is the performance evaluation of the retrieval function. The *precision* and *recall* values of the CBR system retrieval function are calculated according to a threshold value. For this study, the similarity interval [0.6, 1.0] is adopted. This means threshold = 0.6 and cases with global similarity score greater than 60% are retrieved.

In the second phase, the evaluation has been done to judge the system performance on retrieving relevant cases. The *precision* and *recall* are calculated comparing the Business Model Gallery against our retrieval method enhanced with knowledge. For the evaluation, four test cases as query are created. The comparison results between the two systems in terms of *precision* and *recall* are stated in Table 2 where 0.6 is considered as a threshold value. The results from Table 2 indicate that the *average precision* of four test cases for the Business Model Gallery is 81.5%, whereas our enhanced system has better *average precision* of 95%. In the same way, the *average recall* for EIDSS-BM is 95.5%, whereas the Business Model Gallery has a lower value of 46%. On the whole, the results suggest that the enhanced system performs better in terms of *precision* and *recall* in this CBR system. Results demonstrated that CBR provides an effective approach to solving and suggesting BM. Further research in this particular area will continue to investigate more advanced features.

	EIDSS-BM		Business M	lodel Gallery
-	Recall	Precision	Recall	Precision
Problem 1	94%	100%	52%	100%
Problem 2	94%	100%	31%	33%
Problem 3	94%	100%	68%	93%
Problem 4	100%	80%	33%	100%
Average	95.5%	95%	46%	81.5%

Table 2. Experimental results comparing two systems in terms of precision and recall.

8. Conclusions

In this paper, we are interested in the AI contributions to the Sustainable Development through an explainable intelligent system for BM design. Therefore, digital technologies can help designing BMs, increasing productivity, reducing production costs and emissions, decreasing the intensity of production process resources and improving correspondence in markets.

This paper has presented the first version of the CBR-based EIDSS BM. Our approach to the explainable decision support system for BM design is based on ontologies, and makes knowledge assets intelligently accessible to people in organizations, that knowledge resided in the heads of people only. The valuable knowledge of experts in past business models for many case studies is capitalized, which enables reusing existing domain knowledge as well as the transferring and sharing of knowledge. Consequently, applying this paradigm for software tools of business model development will make great promise for supporting business model design and innovation sustainability. We presented our contribution which led to the development of new generation of intelligent decision support systems for Business called "Explainable Intelligent Decision Support System for Business Model" that takes highly accurate decisional guidance into account by integrating explanations for designing and innovating business models. We presented also a domain-independent CBR platform. The development of a new CBR system is done by providing its domain ontology. Our proposed solution can be used for two parallel objectives: to capitalize knowledge about BM, and to provide support to experts and Business students to design, innovate, and validate their BMs. Experts can use any concept or instance of the domain model to describe their cases.

Future work: For future research direction, we plan to develop a Rule-Based Reasoning module to predict the success of early-stage BMs. The SWRL language could be used to express the determinants of success of companies' rules, based on the analysis of classes and properties in the BM ontology model. Furthermore, the ontology development is a time-consuming task, and initial work is done. A significant future manual effort is required to extend the ontology model with knowledge about management and sustainable product, services, and strategies.

Moreover, we want to extend the intelligent system by a module of the Balanced Scorecard for Business model designed from the understanding of the organization's strategy. The indicators measure the performance of each of the choices of each block of the Business Model case. This module of the Balanced Scorecard allows the decision maker to monitor and visualization the performance of the Business Model case.

Author Contributions: Each author (B.H., A.B., A.K. and M.B.) has equally contributed to this publication. Conceptualization, B.H. and A.K.; methodology, B.H. and A.B.; validation, B.H., A.B., A.K. and M.B.; formal analysis, B.H. and A.K.; resources, B.H. and A.B.; data curation, B.H., A.B. and A.K.; writing—original draft preparation, B.H., A.B., A.K. and M.B.; writing—review and editing, B.H., A.B., A.K. and M.B.; visualization, B.H., A.B., A.K. and M.B.; supervision, A.B., A.K. and M.B.; project administration, A.B. and A.K.; funding acquisition, A.B. and A.K. All authors have read and agreed to the published version of the manuscript.

Funding: This work is fully funded by the General Direction of Scientific Research and Technological Development-Algerian Ministry of Higher Education and Scientific Research DGRSDT/MERS.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The data presented in this article are available on request from the corresponding author.

Acknowledgments: The authors would like to thank the anonymous referees for their valuable comments that improved the quality of the manuscript.

Conflicts of Interest: The authors declare no conflict of interest.

Abbreviations

The following abbreviations are used in this manuscript:

BM	Business Model
CBR	Case-Based Reasoning
DSS	Decision Support Systems
BMDT	Business Model Development Tools
IS	Information System
AI	artificial intelligence
XAI	Explainable Artificial Intelligence
SDG	Sustainable Development Goal
SBMO	Strategic Business Model Ontology
KICBR	Knowledge Intensive Case Based Reasoning
ANN	Artificial Neural Networks

- SVM Support Vector Machines
- KMS Knowledge Management Systems
- SBM Sustainable Business Model
- RBR Rule-Based Reasoning
- NLP Natural Language Processing

References

- 1. Malhotra, A.; Melville, N.P.; Watson, R.T. Spurring impactful research on information systems for environmental sustainability. *MIS Q.* **2013**, *37*, 1265–1274. [CrossRef]
- Ripple, W.J.; Wolf, C.; Newsome, T.M.; Galetti, M.; Alamgir, M.; Crist, E.; Mahmoud, M.I.; Laurance, W.F.; 15,364 Scientist Signatories from 184 Countries. World scientists' warning to humanity: A second notice. *BioScience* 2017, 67, 1026–1028. [CrossRef]
- 3. Broman, G.I.; Robèrt, K.H. A framework for strategic sustainable development. J. Clean. Prod. 2017, 140, 17–31. [CrossRef]
- Vidmar, D. Effects of digital technologies on sustainability performance: Business model perspective. In Proceedings of the 32nd Bled eConference Humanizing Technology for a Sustainable Society, Bled, Slovenia, 16–19 June 2019; University of Maribor Press: Maribor, Slovenia, 2019; pp. 1231–1244. [CrossRef]
- Geissdoerfer, M.; Vladimirova, D.; Evans, S. Sustainable business model innovation: A review. J. Clean. Prod. 2018, 198, 401–416. [CrossRef]
- 6. Gutierrez, A.; Serrano, A. Assessing strategic, tactical and operational alignment factors for SMEs: Alignment across the organisation's value chain. *Int. J. Value Chain Manag.* **2008**, *2*, 33–56. [CrossRef]
- 7. Shekhovtsov, A.; Kołodziejczyk, J.; Sałabun, W. Fuzzy model identification using monolithic and structured approaches in decision problems with partially incomplete data. *Symmetry* **2020**, *12*, 1541. [CrossRef]
- 8. Mulliner, E.; Malys, N.; Maliene, V. Comparative analysis of MCDM methods for the assessment of sustainable housing affordability. *Omega* **2016**, *59*, 146–156. [CrossRef]
- 9. Tzeng, G.H.; Chiang, C.H.; Li, C.W. Evaluating intertwined effects in e-learning programs: A novel hybrid MCDM model based on factor analysis and DEMATEL. *Expert Syst. Appl.* **2007**, *32*, 1028–1044. [CrossRef]
- 10. Wang, J.J.; Jing, Y.Y.; Zhang, C.F.; Zhao, J.H. Review on multi-criteria decision analysis aid in sustainable energy decision-making. *Renew. Sustain. Energy Rev.* 2009, 13, 2263–2278. [CrossRef]
- 11. Kizielewicz, B.; Sałabun, W. A new approach to identifying a multi-criteria decision model based on stochastic optimization techniques. *Symmetry* **2020**, *12*, 1551. [CrossRef]
- 12. Ur Rehman, A.; Shekhovtsov, A.; Rehman, N.; Faizi, S.; Sałabun, W. On the Analytic Hierarchy Process Structure in Group Decision-Making Using Incomplete Fuzzy Information with Applications. *Symmetry* **2021**, *13*, 609. [CrossRef]
- 13. Sauter, V.L. Decision Support Systems for Business Intelligence; John Wiley & Sons: Hoboken, NJ, USA, 2014.
- 14. GeSI. Enabling the Low Carbon Economy in the Information Age, 2008; GeSI: London, UK, 2020.
- 15. Global e-Sustainability Initiative. *GeSI SMARTer 2020: The Role of ICT in Driving a Sustainable Future;* Global e-Sustainability Initiative: Brussels, Belgium, 2012.
- 16. Laitner, J.A.S. The energy efficiency benefits and the economic imperative of ICT-enabled systems. In *ICT Innovations for Sustainability*; Springer: Berlin/Heidelberg, Germany, 2015; pp. 37–48.
- 17. Nilsson, N.J. Artificial intelligence prepares for 2001. AI Mag. 1983, 4, 7. [CrossRef]
- 18. Di Vaio, A.; Palladino, R.; Hassan, R.; Escobar, O. Artificial intelligence and business models in the sustainable development goals perspective: A systematic literature review. *J. Bus. Res.* **2020**, *121*, 283–314. [CrossRef]
- 19. Acemoglu, D.; Restrepo, P. 8. Artificial Intelligence, Automation, and Work; University of Chicago Press: Chicago, IL, USA, 2019.
- 20. Norouzzadeh, M.S.; Nguyen, A.; Kosmala, M.; Swanson, A.; Palmer, M.S.; Packer, C.; Clune, J. Automatically identifying, counting, and describing wild animals in camera-trap images with deep learning. *Proc. Natl. Acad. Sci. USA* **2018**, *115*, E5716–E5725. [CrossRef] [PubMed]
- 21. Vinuesa, R.; Azizpour, H.; Leite, I.; Balaam, M.; Dignum, V.; Domisch, S.; Felländer, A.; Langhans, S.D.; Tegmark, M.; Nerini, F.F. The role of artificial intelligence in achieving the Sustainable Development Goals. *Nat. Commun.* **2020**, *11*, 1–10. [CrossRef]
- 22. Gretzel, U.; Sigala, M.; Xiang, Z.; Koo, C. Smart tourism: Foundations and developments. *Electron. Mark.* 2015, 25, 179–188. [CrossRef]
- 23. Joyce, A.; Paquin, R.L. The triple layered business model canvas: A tool to design more sustainable business models. *J. Clean. Prod.* **2016**, 135, 1474–1486. [CrossRef]
- 24. Vaio, A.D.; Boccia, F.; Landriani, L.; Palladino, R. Artificial Intelligence in the Agri-Food System: Rethinking Sustainable Business Models in the COVID-19 Scenario. *Sustainability* **2020**, *12*, 4851. [CrossRef]
- 25. Kwon, M.; Mlecnik, E.; Gruis, V. Business Model Development for Temporary Home Renovation Consultancy Centres: Experiences from European Pop-Ups. *Sustainability* **2021**, *13*, 8450. [CrossRef]
- 26. Schaltegger, S.; Lüdeke-Freund, F.; Hansen, E.G. *Business Cases for Sustainability and the Role of Business Model Innovation: Developing a Conceptual Framework*; Centre for Sustainability Management (CSM), Leuphana Universität Lüneburg: Lüneburg, Germany, 2011.
- 27. Stubbs, W.; Cocklin, C. Conceptualizing a "sustainability business model". Organ. Environ. 2008, 21, 103–127. [CrossRef]

- 28. Pedersen, E.R.G.; Gwozdz, W.; Hvass, K.K. Exploring the relationship between business model innovation, corporate sustainability, and organisational values within the fashion industry. *J. Bus. Ethics* **2018**, *149*, 267–284. [CrossRef]
- Mathues, J.; Desmet, C.; Stein Ivar, S. Tools for Sustainable Business Model Development: A Comparative Case Study on a Non-Profit Organisation. Master's Thesis, Louvain School of Management, Université Catholique de Louvain, Ottignies-Louvainla-Neuve, Belgium, 2019. Available online: http://hdl.handle.net/2078.1/thesis:18874 (accessed on 1 August 2021).
- 30. Kropp, J.; Scheffran, J. Advanced Methods for Decision Making and Risk Management in Sustainability Science; Nova Publishers: Hauppauge, NY, USA, 2007.
- 31. Bouwman, H.; de Vos, H.; Haaker, T. *Mobile Service Innovation and Business Models*; Springer Science & Business Media: Berlin/Heidelberg, Germany, 2008.
- 32. Osterwalder, A. The Business Model Ontology—A Proposition in a Design Science Approach University of Lausanne. Ph.D. Thesis, University of Lausanne, Lausanne, Switzerland, 2004.
- 33. Gordijn, J.; Akkermans, J. Value-based requirements engineering: Exploring innovative e-commerce ideas. *Requir. Eng.* **2003**, *8*, 114–134.
- 34. Sinkovics, N.; Gunaratne, D.; Sinkovics, R.R.; Molina-Castillo, F.J. Sustainable Business Model Innovation: An Umbrella Review. *Sustainability* **2021**, *13*, 7266. [CrossRef]
- 35. López-Nicolás, C.; Ruiz-Nicolás, J.; Mateo-Ortuño, E. Towards Sustainable Innovative Business Models. *Sustainability* **2021**, 13, 5804. [CrossRef]
- 36. Schneider, S.; Spieth, P. Business model innovation: Towards an integrated future research agenda. *Int. J. Innov. Manag.* 2013, 17, 1340001. [CrossRef]
- Szopinski, D.; Schoormann, T.; John, T.; Knackstedt, R.; Kundisch, D. How software can support innovating business models: A taxonomy of functions of business model development tools. In Proceedings of the Association for Information Systems (AIS), Boston, MA, USA, 10–12 August 2017.
- Ebel, P.; Bretschneider, U.; Leimeister, J.M. Leveraging virtual business model innovation: A framework for designing business model development tools. *Inf. Syst. J.* 2016, 26, 519–550. [CrossRef]
- Osterwalder, A.; Pigneur, Y. Designing business models and similar strategic objects: The contribution of IS. J. Assoc. Inf. Syst. 2012, 14, 3. [CrossRef]
- 40. Veit, D.; Clemons, E.; Benlian, A.; Buxmann, P.; Hess, T.; Kundisch, D.; Leimeister, J.M.; Loos, P.; Spann, M. Business models. *Bus. Inf. Syst. Eng.* **2014**, *6*, 45–53. [CrossRef]
- 41. Dellermann, D.; Lipusch, N.; Ebel, P.; Leimeister, J.M. Design principles for a hybrid intelligence decision support system for business model validation. *Electron. Mark.* 2019, 29, 423–441. [CrossRef]
- Majchrzak, A.; Gasser, L. On using artificial intelligence to integrate the design of organizational and process change in US manufacturing. *AI Soc.* 1991, *5*, 321–338. [CrossRef]
- 43. Snihur, Y.; Lamine, W.; Wright, M. Educating engineers to develop new business models: Exploiting entrepreneurial opportunities in technology-based firms. *Technol. Forecast. Soc. Chang.* **2018**, *164*, 119518. [CrossRef]
- 44. Martinsons, M.G.; Davison, R.M. Strategic decision making and support systems: Comparing American, Japanese and Chinese management. *Decis. Support Syst.* 2007, 43, 284–300. [CrossRef]
- 45. McNurlin, B.C.; Sprague, R.H. Information Systems Management in Practice; Prentice Hall PTR: Hoboken, NJ, USA, 2001.
- 46. Zaraté, P.; Liu, S. A new trend for knowledge-based decision support systems design. *Int. J. Inf. Decis. Sci.* 2016, *8*, 305–324. [CrossRef]
- 47. Power, D.J. A Brief History of Decision Support Systems. Volume 3, 2007. Available online: http://www.dssresources.com/ (accessed on 21 June 2021).
- Liu, S.; Delibasic, B.; Butel, L.; Han, X. Sustainable knowledge-based decision support systems (DSS): Perspectives, new challenges and recent advance. *Ind. Manag. Data Syst.* 2017, 117, 1318–1322. [CrossRef]
- 49. Bergmann, R. *Experience Management: Foundations, Development Methodology, and Internet-Based Applications;* Springer: Berlin/Heidelberg, Germany, 2003; Volume 2432.
- 50. Aamodt, A.; Plaza, E. Case-based reasoning: Foundational Issues, Methodological Variations, and System Approaches. *AI Commun.* **1994**, *7*, 39–59. [CrossRef]
- 51. Arcos, J.L.; De Mantaras, R.L.; Serra, X. Saxex: A case-based reasoning system for generating expressive musical performances. *J. New Music Res.* **1998**, 27, 194–210. [CrossRef]
- 52. Lim, G.; Ahn, H.; Lee, H. Formulating strategies for stakeholder management: A case-based reasoning approach. *Expert Syst. Appl.* **2005**, *28*, 831–840. [CrossRef]
- Changchien, S.W.; Lin, M.C. Design and implementation of a case-based reasoning system for marketing plans. *Expert Syst. Appl.* 2005, 28, 43–53. [CrossRef]
- 54. Kolodner, J.L. Understanding creativity: A case-based approach. In *European Workshop on Case-Based Reasoning*; Springer: Berlin/Heidelberg, Germany, 1993; pp. 1–20.
- 55. Shortliffe, E.H.; Buchanan, B.G. Rule-Based Expert Systems: The MYCIN Experiments of the Stanford Heuristic Programming Project; Addison-Wesley Publishing Company: Boston, MA, USA, 1985.
- 56. Sørmo, F.; Cassens, J.; Aamodt, A. Explanation in case-based reasoning-perspectives and goals. *Artif. Intell. Rev.* 2005, 24, 109–143. [CrossRef]

- 57. Ashley, K.D.; Rissland, E.L. Compare and Contrast: A Test of Expertise. AAAI 1987, 87, 273–278.
- Massie, S.; Craw, S.; Wiratunga, N. A visualisation tool to explain case-base reasoning solutions for tablet formulation. In Proceedings of the International Conference on Innovative Techniques and Applications of Artificial Intelligence, Cambridge, UK, 13–15 December 2004; Springer: Berlin/Heidelberg, Germany, 2004; pp. 222–234.
- Al-Debi, M.M.; El-Haddadeh, R.; Avison, D. Defining the business model in the new world of digital business. In Proceedings of the AMCIS 2008 Proceedings, Toronto, ON, Canada, 14–17 August 2008; p. 300.
- 60. Gordijn, J.; Akkermans, H. Designing and evaluating e-business models. IEEE Intell. Syst. 2001, 16, 11–17. [CrossRef]
- Terrenghi, N.; Schwarz, J.; Legner, C.; Eisert, U. Business model management: Current practices, required activities and IT support. In Proceedings of the Internationale Tagung Wirtschaftsinformatik 2017, St. Gallen, Switzerland, 12–15 February 2017.
 Amit, R.; Zott, C. Value creation in e-business. *Strateg. Manag. J.* 2001, *22*, 493–520. [CrossRef]
- 62. Amit, R.; Zott, C. Value creation in e-business. *Strateg. Manag. J.* **2001**, *22*, 493–520. [CrossRef]
- Chesbrough, H. Business model innovation: It's not just about technology anymore. *Strategy Leadersh.* 2007, 35, 12–17. [CrossRef]
 Kshetri, N. Barriers to e-commerce and competitive business models in developing countries: A case study. *Electron. Commer.*
- 64. Kshetri, N. Barriers to e-commerce and competitive business models in developing countries: A case study. *Electron. Commer. Res. Appl.* 2007, *6*, 443–452. [CrossRef]
 65. Commerce and Competitive business models in developing countries: A case study. *Electron. Commerce.*
- 65. Sosna, M.; Trevinyo-Rodríguez, R.N.; Velamuri, S.R. Business model innovation through trial-and-error learning: The Naturhouse case. *Long Range Plan.* **2010**, *43*, 383–407. [CrossRef]
- Kley, F.; Lerch, C.; Dallinger, D. New business models for electric cars—A holistic approach. *Energy Policy* 2011, *39*, 3392–3403. [CrossRef]
- 67. Kinder, T. Emerging e-commerce business models: An analysis of case studies from West Lothian, Scotland. *Eur. J. Innov. Manag.* **2002**, *5*, 130–151. [CrossRef]
- 68. Bouwman, H.; De Reuver, M.; Droes, R.; De Vos, H.; Faber, E.; Fielt, E.; Haaker, T.; Ter Hofte, H.; Van der Steen, M.; Van der Roest, H.; et al. Creating Successful ICT Services. In *Practical Guidelines Based on the STOF Method*; Telematica Instituut: Enschede, The Netherlands, 2008.
- 69. Osterwalder, A.; Pigneur, Y. Business Model Generation: A Handbook for Visionaries, Game Changers, and Challengers; John Wiley & Sons: Hoboken, NJ, USA, 2010.
- 70. Samavi, R.; Yu, E.; Topaloglou, T. Strategic reasoning about business models: A conceptual modeling approach. *Inf. Syst. E-Bus. Manag.* **2009**, *7*, 171–198. [CrossRef]
- 71. Bouwman, H.; Zhengjia, M.; Van der Duin, P.; Limonard, S. A business model for IPTV service: A dynamic framework. *Info* 2008, 10, 22–38. [CrossRef]
- 72. Gassmann, O.; Frankenberger, K.; Csik, M. *The Business Model Navigator: 55 Models that Will Revolutionise Your Business;* Pearson: London, UK, 2014.
- 73. Stampfl, G.; Sniukas, M. Borrow Business Models to Reinvent Your Industry. 2015. Available online: https://innovationmanagement.se/2013/11/04/borrow-business-models-to-reinvent-your-industry/ (accessed on 2 August 2021).
- 74. Abdelkafi, N.; Makhotin, S.; Posselt, T. Business model innovations for electric mobility—What can be learned from existing business model patterns? *Int. J. Innov. Manag.* **2013**, *17*, 1340003. [CrossRef]
- 75. Geissdoerfer, M.; Weerdmeester, R. Managing business model innovation for relocalization in the process and manufacturing industry. *J. Bus. Chem.* **2019**, *16*, 11–25. [CrossRef]
- 76. Haaker, T.; Bouwman, H.; Janssen, W.; de Reuver, M. Business model stress testing: A practical approach to test the robustness of a business model. *Futures* **2017**, *89*, 14–25. [CrossRef]
- 77. Daas, D.; Hurkmans, T.; Overbeek, S.; Bouwman, H. Developing a decision support system for business model design. *Electron. Mark.* 2013, 23, 251–265. [CrossRef]
- 78. Euchner, J.; Ganguly, A. Business model innovation in practice. Res.-Technol. Manag. 2014, 57, 33–39.
- 79. Osterwalder, A.; Pigneur, Y.; Tucci, C.L. Clarifying business models: Origins, present, and future of the concept. *Commun. Assoc. Inf. Syst.* **2005**, *16*, 1. [CrossRef]
- Zoric, J. Connecting business models with service platform designs: Exploiting potential of scenario modeling. *Telemat. Inform.* 2011, 28, 40–54. [CrossRef]
- 81. Pagani, M. Roadmapping 3G mobile TV: Strategic thinking and scenario planning through repeated cross-impact handling. *Technol. Forecast. Soc. Chang.* **2009**, *76*, 382–395. [CrossRef]
- Hamrouni, B.; Korichi, A.; Bourouis, A. IDSS-BM: Intelligent Decision Support System for Business Models. In Proceedings of the 7th International Conference on Software Engineering and New Technologies, Hammamet, Tunisie, 26–28 December 2018; pp. 1–5.
- 83. Benjamins, V.R.; Fensel, D.; Gómez-Pérez, A. Knowledge Management through Ontologies. PAKM 1998, 98, 5.
- 84. Schönborn, J.M.; Althoff, K.D. Explaining an Argumentation. Differences and Structural Analysis as a Foundation to Improve Case-Based Explanation. In *Knowledge Management in Digital Work Environments—State-of-the-Art and Outlook, Proceedings of the* 10th Conference on Professional Knowledge Management (WM-2019), Potsdam, Germany, 18–20 March 2019; Heisig, P., Ed.; Potsdam University of Applied Sciences, FHP: Potsdam, Germany, 2019.
- 85. Swartout, W.R. XPLAIN: A system for creating and explaining expert consulting programs. *Artif. Intell.* **1983**, *21*, 285–325. [CrossRef]
- 86. Davis, R.; Buchanan, B.; Shortliffe, E. Production rules as a representation for a knowledge-based consultation program. *Artif. Intell.* **1977**, *8*, 15–45. [CrossRef]

- 87. Andrews, R.; Diederich, J.; Tickle, A.B. Survey and critique of techniques for extracting rules from trained artificial neural networks. *Knowl.-Based Syst.* **1995**, *8*, 373–389. [CrossRef]
- Núñez, H.; Angulo, C.; Català, A. Rule extraction from support vector machines. In Proceedings of the ESANN 2002, Bruges, Belgium, 24–26 April 2002; pp. 107–112.
- 89. Cramer, H.; Evers, V.; Ramlal, S.; Van Someren, M.; Rutledge, L.; Stash, N.; Aroyo, L.; Wielinga, B. The effects of transparency on trust in and acceptance of a content-based art recommender. *User Model. User-Adapt. Interact.* **2008**, *18*, 455. [CrossRef]
- 90. Herlocker, J.L.; Konstan, J.A.; Riedl, J. Explaining collaborative filtering recommendations. In Proceedings of the 2000 ACM Conference on Computer Supported Cooperative Work, Philadelphia, PA, USA, 2–6 December 2000; pp. 241–250.
- Abdul, A.; Vermeulen, J.; Wang, D.; Lim, B.Y.; Kankanhalli, M. Trends and trajectories for explainable, accountable and intelligible systems: An hci research agenda. In Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems, Montreal, QC, Canada, 21–26 April 2018; pp. 1–18.
- 92. Lacave, C.; Díez, F.J. A review of explanation methods for Bayesian networks. Knowl. Eng. Rev. 2002, 17, 107–127. [CrossRef]
- 93. Confalonieri, R.; del Prado, F.M.; Agramunt, S.; Malagarriga, D.; Faggion, D.; Weyde, T.; Besold, T.R. An Ontology-based Approach to Explaining Artificial Neural Networks. *arXiv* 2019, arXiv:1906.08362.
- Wang, D.; Yang, Q.; Abdul, A.; Lim, B.Y. Designing theory-driven user-centric explainable AI. In Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems, Glasgow, UK, 4–9 May 2019; pp. 1–15.
- 95. MacQueen, J. Some methods for classification and analysis of multivariate observations. In Proceedings of the Fifth Berkeley Symposium on Mathematical Statistics and Probability, Oakland, CA, USA, 1 January 1967; pp. 281–297.
- 96. Quinlan, J.R. C4. 5: Programs for Machine Learning; Elsevier: Amsterdam, The Netherlands, 2014.
- 97. Altman, N.S. An introduction to kernel and nearest-neighbor nonparametric regression. Am. Stat. 1992, 46, 175–185.
- 98. Breese, J.S.; Heckerman, D.; Kadie, C. Empirical analysis of predictive algorithms for collaborative filtering. *arXiv* 2013, arXiv:1301.7363.
- 99. Hotelling, H. Analysis of a complex of statistical variables into principal components. J. Educ. Psychol. 1933, 24, 417. [CrossRef]
- 100. Koren, Y.; Bell, R.; Volinsky, C. Matrix factorization techniques for recommender systems. Computer 2009, 42, 30–37. [CrossRef]
- 101. Gunning, D. *Explainable Artificial Intelligence (Xai)*; Defense Advanced Research Projects Agency (DARPA): Arlington County, VA, USA, 2017; Volume 2.
- 102. van Melle, W.; Shortliffe, E.H.; Buchanan, B.G. EMYCIN: A knowledge engineer's tool for constructing rule-based expert systems. In *Rule-Based Expert Systems: The MYCIN Experiments of the Stanford Heuristic Programming Project*; Infotech Ltd.: Maidenhead, UK, 1984; pp. 302–313.
- 103. Khan, M.J.; Hayat, H.; Awan, I. Hybrid case-base maintenance approach for modeling large scale case-based reasoning systems. *Hum.-Centric Comput. Inf. Sci.* 2019, 9, 9. [CrossRef]
- 104. Varma, A. ICARUS: Design and deployment of a case-based reasoning system for locomotive diagnostics. In *International Conference on Case-Based Reasoning*; Springer: Berlin/Heidelberg, Germany, 1999; pp. 581–596.
- 105. Segovia, J.; Szczepaniak, P.S.; Niedzwiedzinski, M. *E-Commerce and Intelligent Methods*; Physica: Heidelberg, Germany, 2013; Volume 105.
- 106. Park, C.S.; Han, I. A case-based reasoning with the feature weights derived by analytic hierarchy process for bankruptcy prediction. *Expert Syst. Appl.* 2002, 23, 255–264. [CrossRef]
- 107. Schmidt, G. Case-based reasoning for production scheduling. Int. J. Prod. Econ. 1998, 56, 537–546. [CrossRef]
- 108. Chang, H.C.; Dong, L.; Liu, F.; Lu, W.F. Indexing and retrieval in machining process planning using case-based reasoning. *Artif. Intell. Eng.* **2000**, *14*, 1–13. [CrossRef]
- 109. Chiu, C. A case-based customer classification approach for direct marketing. Expert Syst. Appl. 2002, 22, 163–168. [CrossRef]
- 110. Liao, T.; Zhang, Z.; Mount, C. A case-based reasoning system for identifying failure mechanisms. *Eng. Appl. Artif. Intell.* 2000, 13, 199–213. [CrossRef]
- 111. Yang, B.S.; Han, T.; Kim, Y.S. Integration of ART-Kohonen neural network and case-based reasoning for intelligent fault diagnosis. *Expert Syst. Appl.* 2004, 26, 387–395. [CrossRef]
- 112. Hsu, C.I.; Chiu, C.; Hsu, P.L. Predicting information systems outsourcing success using a hierarchical design of case-based reasoning. *Expert Syst. Appl.* **2004**, *26*, 435–441. [CrossRef]
- Haque, B.U.; Belecheanu, R.; Barson, R.J.; Pawar, K.S. Towards the application of case based reasoning to decision-making in concurrent product development (concurrent engineering). In *Applications and Innovations in Intelligent Systems VII*; Springer: Berlin/Heidelberg, Germany, 2000; pp. 81–101.
- Jung, C.; Han, I.; Suh, B. Risk analysis for electronic commerce using case-based reasoning. *Intell. Syst. Account. Financ. Manag.* 1999, 8, 61–73. [CrossRef]
- 115. Noh, J.; Lee, K.; Kim, J.; Lee, J.; Kim, S.H. A case-based reasoning approach to cognitive map-driven tacitknowledge management. *Expert Syst. Appl.* **2000**, *19*, 249–259. [CrossRef]
- 116. Perner, P. Mining sparse and big data by case-based reasoning. Procedia Comput. Sci. 2014, 35, 19–33. [CrossRef]
- 117. Gómez-Pérez, A. Knowledge sharing and reuse. Handb. Appl. Expert Syst. 1998, 10, 1–36.
- Díaz-Agudo, B.; González-Calero, P.A. An architecture for knowledge intensive CBR systems. In Proceedings of the European Workshop on Advances in Case-Based Reasoning, Trento, Italy, 6–9 September 2000; Springer: Berlin/Heidelberg, Germany, 2000; pp. 37–48.

- Kofod-Petersen, A.; Aamodt, A. Case-based reasoning for situation-aware ambient intelligence: A hospital ward evaluation study. In Proceedings of the International Conference on Case-Based Reasoning, Seattle, WA, USA, 20–23 July 2009; Springer: Berlin/Heidelberg, Germany, 2009; pp. 450–464.
- Roth-Berghofer, T.R. Explanations and case-based reasoning: Foundational issues. In European Conference on Case-Based Reasoning; Springer: Berlin/Heidelberg, Germany, 2004; pp. 389–403.
- 121. Cozzi, L.; Turk, D.; Abergel, T.; Bartos, J.; Bellevrat, E.; Bennett, S.; Berly, T.; Bouckaert, S.; Dulac, J.; Alvarez, C.F.; et al. *Digitalization and Energy*; OECD: Paris, France, 2017.
- 122. Fuso Nerini, F.; Slob, A.; Ericsdotter Engström, R.; Trutnevyte, E. A research and innovation agenda for zero-emission European cities. *Sustainability* **2019**, *11*, 1692. [CrossRef]
- 123. Herweijer, C.; Waughray, D. Fourth Industrial Revolution for the Earth Harnessing Artificial Intelligence for the Earth; A Report of PricewaterhouseCoopers (PwC); PricewaterhouseCoopers: London, UK, 2018.
- 124. Vinuesa, R.; de Arévalo, L.F.; Luna, M.; Cachafeiro, H. Simulations and experiments of heat loss from a parabolic trough absorber tube over a range of pressures and gas compositions in the vacuum chamber. J. Renew. Sustain. Energy 2016, 8, 023701. [CrossRef]
- Schneider, S.; Leyer, M. Me or information technology? Adoption of artificial intelligence in the delegation of personal strategic decisions. *Manag. Decis. Econ.* 2019, 40, 223–231. [CrossRef]
- 126. Govindan, K.; Jafarian, A.; Nourbakhsh, V. Designing a sustainable supply chain network integrated with vehicle routing: A comparison of hybrid swarm intelligence metaheuristics. *Comput. Oper. Res.* **2019**, *110*, 220–235. [CrossRef]
- 127. Kahneman, D.; Rosenfield, A.; Gandhi, L.; Blaser, T. Noise. Harvard Bus. Rev. 2016, 38–46.
- 128. Johannessen, J.A.; Olsen, B.; Olaisen, J. Aspects of innovation theory based on knowledge-management. *Int. J. Inf. Manag.* 1999, 19, 121–139. [CrossRef]
- 129. Tsui, E.; Garner, B.J.; Staab, S. The role of artificial intelligence in knowledge management. *Knowl. Based Syst.* **2000**, *13*, 235–239. [CrossRef]
- Hoeschl, H.C.; Barcellos, V. Artificial intelligence and knowledge management. In Proceedings of the IFIP International Conference on Artificial Intelligence in Theory and Practice, Santiago, Chile, 21–24 August 2006; Springer: Berlin/Heidelberg, Germany, 2006; pp. 11–19. [CrossRef]
- 131. Bocken, N.M.; Short, S.W.; Rana, P.; Evans, S. A literature and practice review to develop sustainable business model archetypes. *J. Clean. Prod.* **2014**, *65*, 42–56. [CrossRef]
- 132. Mayor, B.; Toxopeus, H.; McQuaid, S.; Croci, E.; Lucchitta, B.; Reddy, S.E.; Egusquiza, A.; Altamirano, M.A.; Trumbic, T.; Tuerk, A.; et al. State of the Art and Latest Advances in Exploring Business Models for Nature-Based Solutions. *Sustainability* 2021, 13, 7413. [CrossRef]
- 133. Morris, M.; Schindehutte, M.; Allen, J. The entrepreneur's business model: Toward a unified perspective. *J. Bus. Res.* 2005, 58, 726–735. [CrossRef]
- 134. Zilahy, G. Sustainable business models—What do management theories say? Vez.-Bp. Manag. Rev. 2016, 47, 62–72. [CrossRef]
- 135. Herciu, M. Sustainability and Profitability Can Coexist. Improving Business Models. *Fluxos Riscos-Rev. Estud. Soc.* **2019**, *5*, 15–33. 136. Duan, Y.; Edwards, J.S.; Dwivedi, Y.K. Artificial intelligence for decision making in the era of Big Data–evolution, challenges and
- research agenda. Int. J. Inf. Manag. 2019, 48, 63–71. [CrossRef]
- 137. Xu, F.; Liu, X.; Chen, W.; Zhou, C.; Cao, B. Ontology-based method for fault diagnosis of loaders. Sensors 2018, 18, 729. [CrossRef]
- Wu, H.; Zhong, B.; Medjdoub, B.; Xing, X.; Jiao, L. An Ontological Metro Accident Case Retrieval Using CBR and NLP. *Appl. Sci.* 2020, 10, 5298. [CrossRef]
- 139. Abou Assali, A.; Lenne, D.; Debray, B. Case retrieval in ontology-based CBR systems. In Proceedings of the Annual Conference on Artificial Intelligence, Pasadena, CA, USA, 11–17 July 2009; Springer: Berlin/Heidelberg, Germany, 2009; pp. 564–571.
- Recio-Garía, J.A.; Díaz-Agudo, B. Ontology based CBR with jCOLIBRI. In Proceedings of the International Conference on Innovative Techniques and Applications of Artificial Intelligence, Cambridge, UK, 11–13 December 2006; Springer: Berlin/Heidelberg, Germany, 2006; pp. 149–162.
- 141. Baader, F.; Calvanese, D.; McGuinness, D.; Patel-Schneider, P.; Nardi, D. *The Description Logic Handbook: Theory, Implementation and Applications*; Cambridge University Press: Cambridge, UK, 2003.
- 142. Oberle, D. Semantic Management of Middleware; Springer Science & Business Media: Berlin/Heidelberg, Germany, 2006; Volume 1.
- 143. Osterwalder, A. An E-Business Model Ontology for the Creation of New Management Software Tools and IS Requirement Engineering; IFIP DsiAge'2002 Doctoral Consortium: Cork, Ireland, 2002.
- 144. Magretta, J. Why Business Models Matter. Harv. Bus. Rev. 2002, 80, 86–133. [PubMed]
- 145. Rese, M.; Meier, H.; Gesing, J.; Boßlau, M. An ontology of business models for industrial product-service systems. In *The Philosopher's Stone for Sustainability;* Springer: Berlin/Heidelberg, Germany, 2013; pp. 191–196.
- 146. Lambert, S. A conceptual framework for business model research. In Proceedings of the BLED 2008 Proceedings, Bled, Slovenia, 15–18 June 2008; p. 24
- 147. Schank, R.C. Dynamic Memory: A Theory of Reminding and Learning in Computers and People; Cambridge University Press: Cambridge, UK, 1983.
- 148. Schank, R. Explanation Patterns: Understanding Mechanically and Creatively; Psychology Press: Hove, UK, 2013.

- Roth-Berghofer, T.R.; Cassens, J. Mapping goals and kinds of explanations to the knowledge containers of case-based reasoning systems. In Proceedings of the International Conference on Case-Based Reasoning, Chicago, IL, USA, 23–26 August 2005; Springer: Berlin/Heidelberg, Germany, 2005; pp. 451–464.
- 150. Bahls, D.; Roth-Berghofer, T. Explanation support for the case-based reasoning tool mycbr. AAAI 2007, 7, 1844–1845.
- Roth-Berghofer, T.R.; Bahls, D. Explanation capabilities of the open source case-based reasoning tool mycbr. In Proceedings of the Thirteenth UK Workshop on Case-Based Reasoning UKCBR, London, UK, 9 December 2008; pp. 23–34.
- 152. Gennari, J.H.; Musen, M.A.; Fergerson, R.W.; Grosso, W.E.; Crubézy, M.; Eriksson, H.; Noy, N.F.; Tu, S.W. The evolution of Protégé: An environment for knowledge-based systems development. *Int. J. Hum.-Comput. Stud.* **2003**, *58*, 89–123. [CrossRef]
- 153. Bogaerts, S.; Leake, D. A Framework for Rapid and Modular Case-Based Reasoning System Development; Technical Report, TR 617; Computer Science Department: Bloomington, IN, USA, 2005.
- 154. Stahl, A.; Roth-Berghofer, T.R. Rapid prototyping of CBR applications with the open source tool myCBR. In Proceedings of the European Conference on Case-Based Reasoning, Trier, Germany, 1–4 September 2008; Springer: Berlin/Heidelberg, Germany, 2008; pp. 615–629.
- 155. Staab, S.; Studer, R. Handbook on Ontologies; Springer Science & Business Media: Berlin/Heidelberg, Germany, 2010.
- 156. Crunchbase. Business Model Gallery—Crunchbase Company Profile and Funding. 2021. Available online: https://www.crunchbase.com/organization/business-model-gallery (accessed on 2 February 2021).
- 157. Kumar, K.A.; Singh, Y.; Sanyal, S. Hybrid approach using case-based reasoning and rule-based reasoning for domain independent clinical decision support in ICU. *Expert Syst. Appl.* **2009**, *36*, 65–71. [CrossRef]